Author's response to reviews

Title: Applying a private sector capitation model to the management of type 2 diabetes in the South African public sector: a cost-effectiveness analysis.

Authors:

Heinrich C Volmink (hvolmink@gmail.com)
Melanie Y Bertram (bertramm@who.int)
Ruxana Jina (ruxana.jina@wits.ac.za)
Alisha N Wade (alisha.wade@wits.ac.za)
Karen J Hofman (karen.hofman@wits.ac.za)

Version: 5 Date: 12 September 2014

Author's response to reviews: see over
Author's response to reviews

**Title:** Applying a private sector capitation model to the management of type 2 diabetes in the South African public sector: a cost-effectiveness analysis.

**Authors:**

Heinrich C Volmink (hvolmink@gmail.com)
Melanie Y Bertram (bertramm@who.int)
Ruxana Jina (ruxana.jina@wits.ac.za)
Alisha N Wade (alisha.wade@wits.ac.za)
Karen J Hofman (karen.hofman@wits.ac.za)

**Version:** 4
**Date:** 13 August 2014

**Author's response to reviews:** see over
11 August 2014

Ms Clare Partridge  
BMC Health Services Research  
BioMed Central  
238 Grey's Inn Road  
London WC1X 8HB  
United Kingdom

Dear Ms Partridge,

RE: Manuscript submission (BMS: 1440263489106610)

Thank you very much for the positive response to our submission. With reference to the earlier correspondence, please note the following:

1) We have completed the CHEERS checklist, and will upload the document as an additional file.
2) We will upload this covering letter using the ‘Cover letter’ tab, as indicated.
3) Figure titles have been removed from the figure files and are now listed at the end of the main manuscript (page 36).

In addition to the above, we felt it necessary to make a few changes in the main document. These are detailed on the following page.

Yours sincerely,

Karen Hoffman

Professor K Hoffman  
Director, PRICELESS SA  
University of the Witwatersrand, School of Public Health  
Johannesburg, South Africa  
Email: Karen.Hofman@wits.ac.za  
Phone: +2117172083
<table>
<thead>
<tr>
<th>Changes</th>
<th>Location</th>
</tr>
</thead>
<tbody>
<tr>
<td>Added &quot;Secondary data were used to design the model while cost information was obtained from various sources, taking into account public sector billing.&quot;</td>
<td>Page 3, lines 14 and 15.</td>
</tr>
<tr>
<td>Added &quot;... for the DMP against the usual practice model.&quot;</td>
<td>Page 3, line 19</td>
</tr>
<tr>
<td>Added &quot;... bootstrapping analysis.&quot;</td>
<td>Page 3, line 20</td>
</tr>
<tr>
<td>Changed the email address of one of the authors (from <a href="mailto:helmreich.volmink@wits.ac.za">helmreich.volmink@wits.ac.za</a> to <a href="mailto:hvolmink@gmail.com">hvolmink@gmail.com</a>).</td>
<td>Page 2, line 5</td>
</tr>
</tbody>
</table>